

DSFE, 3 (4): 427–451. DOI: 10.3934/DSFE.2023024 Received: 02 September 2023 Revised: 11 December 2023 Accepted: 21 December 2023 Published: 25 December 2023

http://www.aimspress.com/journal/dsfe

Research article

The impact of monetary policy on banks' risk-taking behavior in an emerging economy: The role of Basel II

Changjun Zheng¹, Md Abdul Mannan Khan^{1,2,*}, Mohammad Morshedur Rahman³, Shahed Bin Sadeque³ and Rabiul Islam²

- ¹ School of Management, Huazhong University of Science and Technology, Wuhan, 430074, P.R. China
- ² Department of Accounting and Information Systems, Bangabandhu Sheikh Mujibur Rahman Science and Technology University, Gopalganj - 8100, Bangladesh
- ³ Department of Accounting, University of Chittagong, Chattogram-4331, Bangladesh
- * Correspondence: Email: mannan4612@gmail.com.

Abstract: The main goal of this study is to investigate the impact of monetary policy (MP) on the risk-taking behavior of Bangladeshi banks. It also attempts to examine the role of Basel II in the association between MP and bank risk-taking pre- and post-2010. This study analyzes data from 33 commercial banks in Bangladesh over the 20 years from 2002 to 2021 and uses the two-step system generalized method of moments to address heteroscedasticity and autocorrelation issues. Unlike previous research, this study confirms the significant effect of Basel II on the relationship between MP and banks' risk-taking behavior. The main findings are first that a non-linear U-shaped relationship exists between MP and banks' risk-taking behavior, implying that when bank rate (BR) and cash reserve ratio (CRR) increase, bank credit risk first decreases, then later increases. Second, bank-level characteristics such as liquidity, regulatory capital, and size have a significant effect on risk, whereas bank age has an insignificant effect on risk-taking behavior. Third, MP and Basel II jointly influence risk-taking so that banks take higher risks before implementing Basel II. Overall, thisstudy offers significant practical implications for academics, researchers, and regulators interested in leveraging the findings.

Keywords: monetary policy; risk-taking behavior; Basel II, Bangladesh

JEL Codes: E44, E52, G21, G32

428

Abbreviations: MP: Monetary policy; BR: Bank rate; CRR: Cash reserve ratio; BB: Bangladesh Bank; BRPD: Banking Regulation and Policy Department; RBCA: Risk-based Capital Adequacy; BRPD: Banking Regulation and Policy Department; MCR: Minimum Capital Requirement; SRP: Supervisory Review Process; RWATA: Risk-Weighted Assets to total assets; GMM: Generalized method of moments; OMO: Open Market Operations; SLR: Statutory Liquidity Ratio; PSTM: Panel smooth transition regression model; CAR: Capital adequacy ratio; OLS: Ordinary least squares; NPLTL: Non-performing loan to total loans; GDP: Gross Domestic Product; INF: Inflation; EQTA: Total equity to total assets

1. Introduction

The banking industry is facing numerous challenges due to its rapid expansion. The global financial crises in 2007–2008 exposed weaknesses in the global regulatory framework and banks' risk management practices. The Basel Council on Banking Supervision has developed Basel laws to promote financial stability. To safeguard themselves against credit, operational, and trading book risks, banks are required by Basel to keep a specific amount of capital. To tackle the challenges and make bank capital more risk-absorbent and shock-resilient, the Banking Regulation and Policy Department (BRPD) issued Circular No. 14 on December 30, 2007, to implement Basel II in Bangladesh. As a result, guidelines on Risk-Based Capital Adequacy (RBCA) were introduced on January 1, 2009, in parallel to existing BRPD Circular No. 10 dated November 25, 2002. These guidelines are based on Basel II, which came into force on January 1, 2010, and its successive supplements. Under Basel II, banks in Bangladesh are required to maintain a Minimum Capital Requirement (MCR) or Capital Adequacy Ratio (CAR) of 10% of Risk Weighted Assets (RWA) or Taka 4,000 million in capital, whichever is higher. According to the Supervisory Review Process (SRP), banks are directed to maintain a level of capital greater than the minimum required amount to cover all possible risks in their business. Consequently, it is necessary to consider the effect of Basel on the association between MP and risk-taking. Basel III was implemented very recently; as a result, we have concentrated on Basel II, as it was introduced in 2010 in Bangladesh, and have divided the dataset into two parts, pre- and post-Basel II.

Scholars and policymakers have recently focused more on the association between monetary policy (MP) and bank risk-taking. The main objectives of MP have always been to maintain price stability and control economic cycles. It is an effective macroeconomic tool for achieving such stability, full employment, and economic expansion. These tools are included within the framework of national objectives based on economic stability and development. MP is intended to keep prices stable by controlling inflation based on a low, stable, and steady rate that is compatible with output growth (Acharya et al., 2020; Bekaert et al., 2013; Jung, 2023; Mahrous et al., 2020). Monetary authorities throughout the globe implement MP by taking various approaches, including the operation of open markets, federal fund rates, the ratio of reserve requirements, discounting rates, and monitoring inflation. A low rate of interest may encourage banks to take unjustified risks, but in many nations, this has been disregarded for two primary reasons. First, while inflation control and economic expansion are the monetary authorities' top priorities, they have neglected to emphasize

the importance of financial stability objectives. Second, when establishing MP, financial stability was not given any weight; financial innovation was viewed as the foundation of strong financial institutions (Acharya et al., 2020; Delis et al., 2017).

However, it is interesting to note that, unlike advanced economies, Bangladesh implements MP through a mix of price-based and quantity-based tools. The cash requirement ratio (CRR) and bank rate (BR) are the basis of the MP stance in the country. This is because during the sample period of 2002 to 2021, the monetary authorities widely used these tools to monitor MP effectiveness. The CRR is a particularly effective tool in the Bangladeshi banking sector and was also used in the study by Hussain et al. (2021). Moreover, it is not clear how the CRR promotes financial stability in the banking sector of Bangladesh. To investigate this, this study uses the CRR and BR and compares their effectiveness in a bank risk context. The distinctive features of MP in Bangladesh provide a unique testing ground for the effectiveness of such a tool, which may not be possible in other countries that use single MP instruments.

In this paper, we empirically consider the relationship between MP and banks' willingness to take risks in an emerging country. The prime aim is to examine the effect of MP on the risk-taking intention of banks in Bangladesh. In addition, we investigated the contribution of the Basel II capital accord to the association between MP and banks' risk-taking. This research is unique in that it studies the effects of macro-level data and credit indicators of an emerging country context such as Bangladesh. Consequently, it is essential to assess how the central bank utilizes lending policy to control credit risk by adopting efficient MP. Banking crises have detrimental consequences not only on the major national economies of the nations in which they originate, but, because of the increased interconnection caused by globalization, they also tend to contaminate other economies, harming the financial stability of the entire world. In recent years, greater emphasis has been placed on enhancing the international regulatory framework for banking, which has had a significant effect on national banks' supervisory activities.

The majority of related studies, such as those of Altavilla et al. (2020), Berger and Bouwman (2017), Hussain et al. (2021), Mahrous et al. (2020) and Olivero et al. (2011), have shown how MP and banks' risk-taking behavior are correlated. However, no evidence has been found to support the association between the three issues of MP, Basel II, and risk-taking practices. Moreover, the relationship between Basel II and banks' risk-taking motives remains unclear. Borio and Zhu (2012), Ge et al. (2023), de Moraes and de Mendonça (2019) and Agur and Demertzis (2010) analyzed the relationship between MP and banks' risk-taking. Using a panel dataset from a sample of different Bangladeshi banks from the period 2002 to 2021, we adapted the two-step system generalized method of moments (GMM) developed by Arellano and Bond (1991) and Arellano and Bover (1995). Some studies have demonstrated that MP and bank risk are significantly and favorably correlated. The non-linear link between MP and risk is also supported by evidence from (Brana et al., 2019). This study also shows that the size, capitalization, liquidity, and age of banks have a significant impact on the risk-taking of commercial banks in Bangladesh. In the Bangladeshi context, a possible cause may be the inadequate monitoring of MP, which encourages the banking sector of developing countries to assume excessive risk. However, this can be reduced by implementing adequate supervision of the market mechanism.

To the best of the author's knowledge, no other study has been conducted on Bangladesh or any other country in the region which analyzes the impact of monetary policy on banks' risk-taking behavior with consideration of the effect of Basel II. Moreover, the most closely related studies have been conducted on samples of predominantly large, publicly traded banks. This paper aims to extend the related literature by concentrating on the banking sector consisting of commercial banks that focus primarily on providing loans and advances to the general public or private sector. Its contributions relate to the empirical evidence provided on the association between MP and risk, with implications drawn which have great potential for the authorities in the development of financial stability and suitable policies for improvement of the banking sector in a developing economy such as Bangladesh. The Bangladesh Bank plays an important role as a supervisory authority and is also responsible for the implementation of the country's monetary policy. This study aims to contribute to the ongoing discussion by providing insights into the relationship between monetary policy and the risk-taking behavior of banks in Bangladesh. It is hoped the findings will serve as important input for policymakers, researchers, and academics interested in understanding the impact of policy on such behavior.

The remaining sections are structured in the following manner. Section 2 presents the operational framework of Bangladesh Bank, while Section 3 provides a concise summary of related research. Section 4 explains the methodology, including the data, sampling procedures, variables, econometric model and estimation method. Section 5 outlines the empirical findings and discusses the effect of MP on banks' risk-taking practices, together with the outcomes of the robustness tests conducted. Finally, Section 6 presents the conclusion, suggestions and recommendations.

2. Operational framework of Bangladesh Bank (BB)

Bangladesh Bank (BB) acts as the Central Bank of Bangladesh and was established on December 16, 1971 through the enactment of Bangladesh Bank Order 1972–President's Order No. 127 of 1972 (Amended in 2003). In Strategic Plan (2010–2014), the vision of BB was stated as "To develop continually as a forward looking central bank with competent and committed professionals of high ethical standards, conducting monetary management and financial sector supervision to maintain price stability and financial system robustness, supporting rapid broad-based inclusive economic growth, employment generation and poverty eradication in Bangladesh". BB has been undertaking its principal task of formulating and implementing MP to manage the monetary and credit system of the country, with a view to stabilizing domestic monetary value and maintaining a competitive external par value of the Taka in order to foster the growth and development of the country's productive resources. BB designs its MP stance and monetary and credit programs based on macroeconomic updates and outlooks both at home and overseas. The main objectives of the monetary policy of Bangladesh Bank are: (a) price stability, both internal and external; (b) sustainable growth and development; (c) high employment; (d) economic and efficient use of resources; and (e) stability of the financial and payment system. The tools and instruments employed in the implementation of monetary policy in Bangladesh are the bank rate, open market operations (OMO), repurchase agreements (Repo) and reverse repo, and statutory reserve requirements (SLR and CRR).

In the institutional context of Bangladesh, reserve requirements refer to the amount of funds that banks must hold with the central bank as a percentage of their total deposits. The reserve requirement is an MP tool that central banks use to influence money supply and credit conditions in the economy. In Bangladesh, the regulatory authority responsible for formulating and implementing MP is the country's central bank. The BB formulates its MP to achieve price stability, promote sustainable economic growth, and maintain a stable external sector. Reserve requirements are one of the tools used to achieve these objectives. The BB Order in 1972 and subsequent amendments outline the legal framework governing reserve requirements in Bangladesh. The central bank has the authority to set reserve ratios and specify the types of deposits subject to reserve requirements.

Bangladesh employs both statutory and cash reserve requirements. Banks maintain statutory reserve requirements (SRR) in the form of liquid assets, and the BB determines the required ratio. Cash reserve requirements (CRR) refer to the percentage of total deposits banks must hold in cash with the central bank. Banks are required to calculate their reserve requirements regularly based on the prescribed ratios. The reserve amount can be maintained as cash or approved liquid assets, and banks report their compliance to the BB. Reserve requirements in Bangladesh are used to manage liquidity in the banking system, influence money supply, and achieve broader MP objectives. By adjusting the reserve ratios, the central bank can impact the lending capacity of banks and control inflationary pressures. The effectiveness of reserve requirements in influencing monetary policy transmission depends on factors such as the banking system's liquidity position, interest rate dynamics, and the overall economic environment. The central bank may adjust reserve requirements to align with its MP stance.

Reserve requirements are often used with other MP tools, such as open market operations and policy interest rates. The coordination of these tools allows the central bank to implement a comprehensive MP strategy. The BB periodically reviews the economic conditions and the effectiveness of its MP tools. Based on the review, adjustments to reserve requirements may be made to address emerging challenges or to support economic stability. Understanding the institutional context of reserve requirements in Bangladesh involves consideration of the legal framework, regulatory authority, objectives, impact on monetary policy, and coordination with other policy tools. It reflects the broader efforts of the central bank to maintain financial stability and promote sustainable economic growth (Bangladesh Bank 2022).

3. Literature review

This section specifically examines the various attempts in related research to measure and interpret the MP and risk-taking of financial organizations, together with their influencing factors. However, the interrelationship between MP, risk-taking, and Basel II is receiving increased attention; therefore, studies are highlighted that deal with the relationship between MP and risk-taking, Basel II and risk-taking, and MP and Basel II.

3.1. MP and bank risk-taking

Some studies have examined the effects of MP on banks' willingness to take risks. According to Brana et al. (2019), Chen et al. (2017), de Moraes et al. (2016), Djatche (2019), Fu and Luo (2021), Geng and Zhai (2013) and Wu et al. (2022), loosening MP encourages banks to take more risks. Consequently, there is a negative relationship between MP and bank credit risk. However, Brana et al. (2019) found that the association between the reserve requirement ratio, interest rates and bank risk was nonlinear. They also found that there is a positive relationship between interest rates and return on equity. The nonlinear association between MP and bank risk has also been examined, and it

was found that time and the difference between the current and benchmark interest rates were crucial in establishing the link between MP and banks' risk-taking behavior.

On the other hand, it has also been shown that the MP trend is positively correlated with bank credit risk. Mahrous et al. (2020), for instance, evaluated the impact of MP on the risk-taking of banks in Thailand by employing the inflation targeting procedure rate and quarterly balance sheet data of 13 banks that had operated in the country in the previous ten years. According to their study, there was a positive relationship between the policy rate and the risk-taking of banks, and that exceptionally lengthy, low rates of interest also contributed to increased risk-taking.

Hussain et al. (2021) used RRR and BMLR as MP stances for 67 Chinese commercial banks that were in operation between 2000 and 2012; however, our analysis differs from theirs in a number of aspects. We have shown the CRR and BR to illustrate Bangladesh's MP stance. In an empirical investigation of the risk-taking channel, MP is the key variable and of the greatest importance. A few studies (Bashir et al., 2020; Bui et al., 2021; Dell'Ariccia et al., 2017; Demsetz & Strahan, 1997; Hussain et al., 2021; Naqvi & Pungaliya, 2023; Nguyen & Boateng, 2015; Rahman et al., 2015) have examined the effect of bank-level characteristics (such as size, leverage liquidity, age, and capitalization) on the risk channel. Hussain et al. (2021) considered how market structure, stock market return, adverse external financial positions, and excessive lending affected the willingness to take risks.

3.2. Basel II and monetary policy

Basel II affects the activity of banks and other financial institutions by mandating capital requirements for bank exposures. When a bank extends credit, it incurs a portfolio risk (Dell'Ariccia et al., 2017; Malovaná et al., 2019; Matthys et al., 2020; Ngambou Djatche, 2022). Al-Tamimi (2008) outlines the main findings of a survey of Asian-Pacific nations, which aimed to determine the stage at which Asia-Pacific institutions were in their Basel II preparations. Blunden (2005) investigated the Basel II accord from the perspective of global consumers and claimed that the Revised Structure would provide consumers with more information rather than the banking decision. In addition, Blunden mentions that a large number of regulators were endeavoring to comply with Basel II by the internationally proposed December 2006 deadline.

Some studies have investigated the effect of MP on banks' risk-taking under the Basel Accord. When a central bank raises the federal funds rate, the deposit rate rises. Therefore, the bank keeps less total capital for the following period, as the likelihood that the capital limit will bind next period has decreased. In this instance, the bank decides to keep less overall capital, increasing the likelihood that its loans will be curtailed in the following quarter. Therefore, a tightening of MP diminishes the value of retained capital and reduces a bank's future loan-making potential. This results in a "bank capital accelerator effect", separate from the borrower-side financial accelerator. Banks fully expect that MP will be in force permanently, so this drop in overall capital will last for a considerable period. Moreover, the higher the persistence of MP, the greater the future impact of interest rate rises on banks' total capital. This reduces bank profits, an effect which is more pronounced when banks have limited capital. Therefore, banks' value decreases when the principal bank raises the interest rate. In a fully defined model of general stability, the fall in bank prominence resulting from a contractionary MP would lead to a decline in economic activity through the conventional pathways described by (Bernanke, 2000; Chami & Cosimano, 2010). Such a decline in economic activity results in a further

reduction in future offerings by banks, as they discover an additional cause to reduce total capital at present and in the future. As stated, this is because the possibility that the restriction in capital will bind in the future has decreased. The change in total capital leads to additional losses in banks' profitability and market value. Consequently, the two typical consequences of MP, a rise in interest rates and a decline in economic action, would be magnified by the effect of bank capital.

3.3. Basel II and bank risk-taking

There is a close relationship between Basel II and banks' risk-taking behavior. In order to protect themselves from impending uncertainty or loss, all European banks already keep the same level of statutory capital (i.e., a minimum of 8% of risk-weighted total assets) (Čihák & Schaeck, 2007). In accordance with Basel-II, Bangladeshi banks must keep the Minimum Capital Requirement (MCR) at 10% of all risk-weighted assets (RWA) or Tk. 4.0 billion in capital, whichever is greater. The ratio of adequate capital varies among banks in a given nation based on their size, ownership, or efficacy, despite their compliance with the same capital law. In conformity with Basel II, the risk-weighted assets to total assets (RWATA) ratio accurately predicts the capital adequacy ratio. Risk-weighted assets are all the assets minus loans and advances of banks, market-valued government securities, and cash. A higher ratio indicates a greater need for CAR, which increases risk-taking (Avery & Berger, 1991; Gropp & Heider, 2007; Naceur et al., 2018; Ogunmola et al., 2022). A study conducted by Gondwe et al. (2022) demonstrates that while regulations relating to supervisory authority (Pillar 2) and activity constraints enhance liquidity risk, laws pertaining to capital (Pillar 1) and market discipline (Pillar 3) are advantageous in minimizing solvency risk (i.e. reducing bank stability).

Some studies have considered how Basel II restrictions affect conventional banks. Theoretically, Basel II guidelines can encourage sophisticated banks to focus on low-risk defaulters and unsophisticated ones to focus on high-risk borrowers, as demonstrated by (Altunbas et al., 2012; Hakenes & Schnabel, 2011; Sarkar & Sensarma, 2019), who show that small banks are encouraged to take greater risks due to the competitive advantage of larger ones. Studies have supported the idea that capital requirements encourage banks to maintain more conservative asset ranges (Furlong & Keeley, 1989) and others that support the opposite idea regarding the effect of capital necessities on the risk-taking of banks in general (Chami & Cosimano, 2010; Gopalakrishnan et al., 2021; VanHoose, 2008). In addition, several studies have shown, in a broader context, that raising capital requirements results in a decrease in bank lending (Aiyar et al., 2014; Rubio & Carrasco-Gallego, 2016; Veeramoothoo & Hammoudeh, 2022).

Therefore, it is clear that there are positive and negative relationships between monetary policy and bank risk-taking in different economies. This study will re-explore these relationships with consideration of the Basel II effect. The nonlinear relationship between them will also be tested. To the best of the author's knowledge, this is the first study to examine the relationship between monetary policy and banks' risk-taking behavior together with the effect of Basel II using the twostep generalized method of moments (GMM) panel estimator.

4. Methodology

Two options are available for GMM estimation: system GMM and difference GMM. We selected difference GMM by Arellano and Bond (1991), based on the (Blundell et al., 2001)

technique for choosing between the two approaches. The pooled OLS approach and fixed effect method should be used to estimate the auto-regressive model. The fixed effect estimation is a lowerbound estimation, whereas the aggregated OLS estimation for the lagged dependent variable is an upper-bound one. If the difference GMM estimation is too near to, or less than, the fixed effect estimation, indicating that the latter is skewed downward due to insufficient instrumentation, system GMM should be considered. The system GMM is effective in this circumstance, as argued by (Blundell et al., 2001; Bond et al., 2001). Similar techniques were employed by (Özşuca & Akbostancı, 2016), who used a similar estimation to empirically evaluate the risk-taking channel in Turkey. Other research has used difference GMM (Chen & Kao, 2014; Hussain et al., 2021; Kabundi & De Simone, 2020). In our investigation, we used difference GMM in accordance with the methodology.

4.1. Data

This study is mostly based on information originating from three secondary sources: Bangladesh Bank's (BB) website, the annual published reports of 33 Bangladeshi scheduled commercial banks, and the World Bank's World Development Indicators. Accounting measures were used for banks' risk-taking. Risk measures and bank-level control variables were collected from the annual published reports of the banks, MP variables from the BB website, and macro-economic control variables from the World Bank's World Development Indicators.

4.2. Sampling technique

There are currently 61 scheduled banks operating in Bangladesh, comprising six commercial banks run by the government and 43 private commercial banks (PCBs), of which 31 are based on interest and eight on Islamic Shari'ah law. There are also nine foreign commercial banks (FCBs) and three specialized banks that do business in Bangladesh. 16 of the PCBs were omitted because they were recently established. We also excluded the nine FCBs and three specialized banks due to data inaccessibility. Therefore, the sample consisted of 33 Bangladeshi commercial banks. A period of 20 years from 2002 to 2021 was covered. An unbalanced panel of 528 bank-level observations was formed due to the fact that data for all institutions was not available for all the years.

4.3. Variable definitions

4.3.1. Main variables

Risk-taking, MP and Basel II were chosen as the three key variables based on the study aims. Two risk variables were utilized: (i) credit risk, as defined by the ratio of nonperforming loans to total loans (NPLTL); and (ii) stability for insolvency risk, an opposite proxy evaluated by the Z-score (LNZ). In their empirical studies, Hussain et al. (2021) and Zheng et al. (2017) make extensive use of both of these variables. A greater NPLTL value indicates a greater credit risk (Changjun Zheng et al., 2017), while a higher Z-score suggests stronger stability (Hussain et al., 2021). The cash requirement ratio (CRR) and bank rate (BR) were used as proxies to measure MP. In addition, to determine the impact of Basel II on the risk-taking of banks in Bangladesh, we adopted a proxy

variable named Basel II, measured by a dummy variable, with the pre-2010 (dummy) variable equal to one, and zero otherwise.

4.3.2. Bank level control variables

Based on previous research, we incorporated a variety of bank-specific control factors into the econometric parameters, including size, liquidity, capitalization, and age, which are supposed to have an impact on the association between MP and banks' risk-taking. Bank size was determined by the natural logarithm of total assets, liquidity as the sum of all loans divided by the sum of all deposits, and the capital adequacy ratio by the ratio of regulatory capital to risk-weighted assets. The age of the bank was the period from its inception to a particular year.

4.3.3. Macroeconomic control variables

By Mateev et al. (2021) and Aiyar et al. (2016), we also took into account two macroeconomic factors that could affect the endogenous variables: the growth rate of gross domestic product (GGDP) and the annual rate of inflation (INFR). Table 1 shows a list of the variables including their short form, and the measures used in the study.

Risk Measures Risk Credit Risk NPL Non-performing loans as a percentage of total loans (Akhter, 2023; ElBannan, 2015; C Zheng et al., 2017) Stability LNZ= Z- Z -Score = (CAR + ROA) /SD (ROA) (Haque, 2019; Moudud-Ul-Huq, 2021; C Score Here, CAR = Capital adequacy ratio, ROA = Return of asset and SD (ROA) = standard deviation of Return on asset There are also a construction of the standard deviation of Return on asset	Variables	Symbol	Measures	Source/References		
RiskCredit RiskNPLNon-performing loans as a percentage of total loans(Akhter, 2023; ElBannan, 2015; C Zheng et al., 2017)StabilityLNZ=Z- ScoreZ -Score = (CAR + ROA) /SD (ROA) Here, CAR = Capital adequacy ratio, ROA = Return of asset and SD (ROA) = standard deviation of Return on asset(Hussein et al., 2017)MoneterrypolicyCPRCash reserve ratio Cash reserve ratio(Hussein et al., 2021; Day et al., 2022)	Risk Measures					
StabilityLNZ=Z -Z -Score = (CAR + ROA) /SD (ROA) Here, CAR = Capital adequacy ratio, ROA = Return of asset and SD (ROA) = standard deviation of Return on asset(Haque, 2019; Moudud-Ul-Huq, 2021; C Zheng et al., 2017)MoneteryPolicyCPRCash reserve ratio (Cash reserve ratio)(Hussein et al., 2021; Day et al., 2022)	Risk Credit Risk NPL Stability LNZ= Z- Score		Non-performing loans as a percentage of total loans	(Akhter, 2023; ElBannan, 2015; C Zheng et al., 2017)		
Monotory policy CDD Cash records ratio (Hussoin at al. 2021; Deviat al. 2022)			Z -Score = $(CAR + ROA) / SD (ROA)$ Here, CAR = Capital adequacy ratio, ROA = Return of asset and SD (ROA) = standard deviation of Return on asset	(Haque, 2019; Moudud-Ul-Huq, 2021; C Zheng et al., 2017)		
(nussain et al., 2021; Koy et al., 2025)	Monetary policy	CRR	Cash reserve ratio	(Hussain et al., 2021; Roy et al., 2023)		
BR Used as bank/short-term interest rate (Anwar et al., 2023; Khan et al., 2016)	(MP)	BR	Used as bank/short-term interest rate	(Anwar et al., 2023; Khan et al., 2016)		
Bank-level variables	Bank-level variables					
Basel II BASEL A dummy equal to 1 for a period before Author calculation 2010 or 2010 and 0 otherwise.	Basel II	BASEL	A dummy equal to 1 for a period before 2010 or 2010 and 0 otherwise.	Author calculation		
Bank size SIZE Total assets as a natural logarithm (Mohammed et al., 2015; Moudud-Ul- Huq, 2021; Ridwan & Mayapada, 2022)	Bank size SIZE		Total assets as a natural logarithm	(Mohammed et al., 2015; Moudud-Ul- Huq, 2021; Ridwan & Mayapada, 2022)		
Liquidity LIQ Total loan & advances to Total deposits (Das Gupta et al., 2021; Khan et al., 2016)	Liquidity LIQ		Total loan & advances to Total deposits	(Das Gupta et al., 2021; Khan et al., 2016)		
Capital adequacy CAR The ratio of regulatory capital to risk- (Ridwan & Mayapada, 2022) weighted assets is also known as the ratio of capital adequacy ratio.	Capital adequacy	CAR	The ratio of regulatory capital to risk- weighted assets is also known as the ratio of capital adequacy ratio.	(Ridwan & Mayapada, 2022)		
Bank age AGE The period from inception to a particular Authors' calculation, (Siddique et al., year 2021)	Bank age	AGE	The period from inception to a particular year	Authors' calculation, (Siddique et al., 2021)		
Macro-economic Variables	Macro-economic Variables		-			
Growth in GDP GDP Annual growth in real Gross Domestic (Akhter, 2023; Rahman et al., 2015) Product	Growth in GDP GDP		Annual growth in real Gross Domestic Product	c (Akhter, 2023; Rahman et al., 2015)		
Inflation rate INFR Annual rate of inflation (Akhter, 2023; Khan, 2022)	Inflation rate	INFR	Annual rate of inflation	(Akhter, 2023; Khan, 2022)		

Table 1. Definition of variables and data sources

Source: Author's compilation using the mentioned sources/references.

In line with the approach of Saif-Alyousfi and Saha (2021) and Hussain et al. (2021), the twostep system generalized method of moments (GMM) developed by Arellano and Bond (1991) and Arellano and Bover (1995) was employed. The aim of employing this approach was to address heteroscedasticity (White test for heteroscedasticity) and autocorrelation (Berusch-Godfrey LM test for autocorrelation) concerns in the data. The accuracy of the model was verified by two standard tests, AR 1 and AR 2, and the Hansen test.

4.3.4. Econometric model

In accordance with the purpose of the study, we employed an empirical econometric model applied in other recent studies, including those of Mateev et al. (2023) and Hussain et al. (2021), and with only minor modifications. Two regressions were used to investigate the various aspects of MP and the risk-taking intention of banks with the help of Basel II. The following regression is displayed by our base model:

$$Y_{i,t} = \beta_0 + \beta_1 Y_{i,t-1} + \beta_2 M P_{i,t} + \beta_3 SIZE_{i,t} + \beta_4 LIQ_{i,t} + \beta_5 AGE_{i,t} + \beta_6 CAR_{i,t} + \beta_7 GDP_t + \beta_8 INFR_t + \epsilon_{i,t}(1)$$

This equation represents the cross-sectional dimensions across banks and time by subscripts i and t. Two proxies were used to identify the dependent variable (Y): nonperforming lending (NPL) and the bank Z-score. The independent variables were derived from the lagged dependent variable, MP, which is represented by two proxies (BR and CRR). The regression model also includes four bank-level variables, bank size, capitalization, liquidity, and age, to investigate the potential effects of bank characteristics on risk. GDP is the real growth rate of gross domestic product, and INFR is the annual rate of inflation. The baseline model was augmented by incorporating Basel II, the effect of bank-level variables on risk-taking behavior, and the effect of external liquidity constraints.

Now, the extension incorporates the impact of Basel II on bank risk-taking. The paradigm presented in Equation 1 can be further developed as follows:

$$Y_{i,t} = \beta_0 + \beta_1 Y_{i,t-1} + \beta_2 M P_{i,t} + \beta_3 SIZE_{i,t} + \beta_4 LIQ_{i,t} + \beta_5 AGE_{i,t} + \beta_6 CAR_{i,t} + \beta_7 GDP_t + \beta_8 INFR_t + \beta_9 BASEL_t + \epsilon_{i,t}$$
(2)

To account for non-linearity between MP and bank risk, and to show the joint effect of MP and Basel II, equations 1 and 2 were further expanded:

$$Y_{i,t} = \beta_0 + \beta_1 Y_{i,t-1} + \beta_2 M P_{i,t} + \beta_3 SIZE_{i,t} + \beta_4 LIQ_{i,t} + \beta_5 AGE_{i,t} + \beta_6 CAR_{i,t} + \beta_7 GDP_t + \beta_8 INFR_t + \beta_9 BASEL_t + \beta_{10} BR_{i,t}^2 + \beta_{11} CRR_{i,t}^2 + \beta_{12} BASEL_t * BR_{i,t} + \beta_{13} BASEL_t * CRR_{i,t} + \epsilon_{i,t}$$
(3)

5. Results and discussion

In this section, a comprehensive overview of the statistics is provided in Table 2, together with the findings of the multicollinearity test in Tables 3 and 4. The experiential results in Tables 5 and 6 were obtained using the two-step difference GMM panel estimator. A Hansen over-identification test was also performed to confirm the credibility of the instruments employed. The applied models are scrutinized by AR (2), which tested the null hypothesis that error terms were not serially correlated. The AR (2) results for all the models used in the study show no second-order serial correlation, as indicated by the p-values.

5.1. Descriptive statistics

Table 2 shows descriptive statistics for the sample of 33 commercial banks operating in Bangladesh, comprising 561 observationsfor the main variables, bank-level variables, and macroeconomic variables. The focus here is primarily on the main variables. Bank risk (measured by NPL) has a mean value of 0.0785752, a standard deviation (SD) of 0.1144728, and a range from 0.0009591 to 1.406547. This indicates that there is a significant degree of variation between banks in terms of non-performing loans. Similarly, the Z-score has a mean value of 1.665862, aSD of 0.641135, and a range of -2.27917 to 3.45657. This variation shows that the risk level differs significantly amongst the banks. The mean values of the MP instruments (cash reserve ratio and bank rates) are 0.0537344 and 4.9110873, respectively. The fact that the standard deviations for the MP instruments are 0.0085567 (CRR) and 0.3574305 (BR) indicates that there are minimal variances in the analysis, which indicates that the MP instruments experienced only minor changes during the study period. However, the lowest values are 0.04 (CRR) and 4.00 (BR), indicating that the MP rates were substantially decreased. The highest levels for both rates, 0.065 (CRR) and 6.00 (BR), indicate that monetary policy rates during the study period were not exceptionally high. The descriptive study demonstrates that during this period, MP rates were relaxed. Before more extensive testing, it may be inappropriate to assert that flexible MP encourages banks to assume excessive risk, despite the fact that the descriptive statistics suggest permissive policy rates.

Variables	No. of obs.	Mean	Std. Dev	Min	Max			
NPL	561	0.078575	0.114472	0.00095	1.40654			
LNZ	561	1.665862	0.641135	-2.27917	3.45657			
BR	561	4.910873	0.357430	4.00000	6.00000			
CRR	561	0.053734	0.008556	0.04000	0.06500			
SIZE	561	11.78162	1.082483	9.07787	14.3077			
CAR	561	10.51320	10.98609	-108.490	22.3200			
LIQ	561	0.851916	0.130478	0.09717	1.33007			
AGE	561	21.85740	10.50447	3.00000	50.0000			
GDP	561	6.197624	1.06443	3.44802	7.88190			
INFR	561	6.785061	1.68487	3.33256	11.3951			
Source: Authors' calculation.								

Table 2. Descriptive statis	stics.
------------------------------------	--------

Notes: Max = Maximum Value, Min = Minimum Value, S.D. = Standard Deviation, and Obs. = Observations.

5.2. Panel Test of stationarity

The unit root test is an important step in determining the stationarity of the variables used to perform the panel data analysis. It helps to identify whether the variables are stationary or not, which is important for accurate analysis and modeling. The panel unit root tests are conducted with Fishertype test statistics (Table 3). The null hypothesis being tested is that the series contains a unit root and that the null hypothesis is rejected in each case at the 5 percent level, and in 6 out of 10 cases easily rejected at the 1 percent level. Given our results, we reject this hypothesis. From the Table 3 it is seen that the value for these test statistics are (269.3086, 180.033, 162.3282, 121.9412, 284.9496, 144.8984, 108.2052, 104.2391, 469.5710 and 309.8023), and in the next column we see the p-value.

This means there are no unit roots in our panels under the given test conditions. Therefore, we can proceed with the two-step difference GMM.

Variables	Statistics	Probability	
NPL	269.3086	0.0000	
LNZ	180.0333	0.0411	
BR	162.3282	0.0320	
CRR	121.9412	0.0300	
SIZE	284.9496	0.0000	
CAR	144.8984	0.0000	
LIQ	108.2052	0.0008	
AGE	104.2391	0.0251	
GDP	469.5710	0.0000	
INFR	309.8023	0.0000	

Table 3. Panel test of stationarity.

5.3. Correlation analysis

Considering Pearson's correlation coefficient, Table 4 presents the correlation matrix. Correlation analysis seeks to determine whether or not multicollinearity is a problem. (Kennedy, 2008) asserts that a correlation problem occurs when the correlation coefficient between two independent variables exceeds 0.70. As can be seen in Table 3, SIZE and AGE have the strongest correlation (Pearson's correlation = 0.58). Multicollinearity is therefore not a problem in our research.

Table 4. Pearson's correlation coefficier	its.
---	------

		1	2	3	4	5	6	7	8	9	10	11
1	NPL	1.0										
	TL											
2	LNZ	-0.04	1.0									
3	SIZE	-0.08	0.19***	1.0								
4	CAR	-0.66^{***}	0.11***	0.27***	1.0							
5	LIQ	-0.28***	-0.02	0.04	0.07	1.0						
6	AGE	0.34***	0.13***	0.58***	-0.08*	-0.28 **	1.0					
					*	*						
7	BR	0.01	-0.25 **	-0.43**	-0.09*	-0.15^{**}	-0.28 * *	-0.32**	1.0			
			*	*	*	*	*	*				
8	CRR	-0.03	-0.09^{**}	0.29***	-0.09*	-0.03**	0.13***	0.50***	0.39**	1.0		
					*	*			*			
9	GDP	-0.08	-0.21**	-0.10**	-0.08*	-0.021	-0.12**	-0.07*	0.17**	0.47***	1.0	
			*	*	*		*		*			
10	INFR	-0.12***	-0.19**	-0.26**	-0.05	-0.04	-0.22 **	-0.26**	0.11**	0.10***	0.56***	1.0
			*	*			*	*	*			

Notes: The total number of observations are 561. *, **, *** denote significance at p < 0.10, p < 0.05 and p < 0.01 (2-tailed) respectively. NPL = Non-performing loans to total loans; LNZ = the return on assets ratio (ROA) plus the CAR divided by the SD of ROA defined as a Z-score is another endogenous variable; CAR = Capital adequacy ratio; ROA = Return on assets; LIQ = Liquidity; AGE = Banks' age from the inception of a bank; BASEL = Basel II; BR = Bank rates; CRR = Cash reserve ratio; GDP = Growth in gross domestic products; INFR = Consumer price index.

The table also shows the correlation between the endogeneous variables (NPLTL, LNZ, BR, and CRR) and the association between the endogeneous, bank level and micro-economic variables. Pearson's correlation co-efficient shows that the connection between MP [both proxies] and risk is negative and significant when risk is measured by LNZ and NPLTL, except BR with NPLTL.

Among the bank level variables, SIZE, CAR, and LIQ have negative impact on NPLTL and a positive impact on LNZ, while AGE has a positive effect on risk [both proxies]. All the bank level control variables have negative impact on BR and CRR, but the association between SIZE, AGE, and CRR is positive. Both macro-level variables (GDP and INFR) have an adverse effect on bank risk. Table 4 also shows that there is a correlation that is very low (less than 0.3) and that some correlations are ambiguous, which suggests that there is a need for regression analysis to have a clear identification of the association among the variables.

5.4. Regression analysis

This section analyzes the above-described equations in which RISK, MP, and CAR are endogenous variables. In this study, the Hausman test was used to observe whether or not simultaneity existed. From the p-value of the Hausman test, we can reject the null hypothesis of no simultaneity among the three variables. On the basis of the p-value of the autocorrelation test, the null hypothesis of no autocorrelation can also be rejected. The null hypothesis of homoscedasticity is rejected at a 5% level of significance when a White test (White, 1980) is employed to investigate cross-sectional heteroscedasticity. Hence, the application of ordinary least squares (OLS) estimation is inappropriate in this situation. To know about endogeneity, autocorrelation, and heteroskedasticity, we relied on GMM estimation. Of the two forms of GMM estimation, thistest supports the application of two-step difference GMM. The following section is divided into two parts for simplicity of discussion. The first part illustrates the effect of MP and Basel II on bank risk-taking as measured by credit risk, while the second part focuses on the effect of MP and Basel II on bank stability, as measured by the Z-score.

5.4.1. The influence of MP and Basel II on risk-taking behavior when credit risk is considered

Regarding the impact of MP and Basel II on bank risk, Table 5 shows the results of the eight regression models. The first two models (M1-M2) describe the base line model, and the second six (M3-M8) the extended model. In model M1, the MP proxy is BR, while in model M2 it is CRR. Basel II is included in models M3 and M4. With reference to Table 5, it can be said that MP, as measured by both BR and CRR, has a significant adverse impact on banks' risk-taking. This suggests that when BR increases, banks' credit risk decreases. The same conclusion can be drawn when CRR increases. A possible reason for this may be the reduction in disbursement due to the increase in BR and CRR, with NPL decreasing as a result. These findings are consistent with those of (de Moraes & de Mendonça, 2019), but contrary to the findings of (Mahrous et al., 2020). These results indicate that a 1% in rise in BR would reduce credit risk by 9.3% for Model 1 and 7.7% for Model 4. It is also indicated that a 1% increase in CRR would reduce credit risk by 92.9% for Model 2 and 54.4% for Model 3. The findings for M1 and M2 are different from those of M3 and M4, which is because of the inclusion of Basel in M3 and M4. The impact of the Basel II regulations on credit risk is negative and highly significant, which suggests that the banks is taking lower credit risk than they were in the period before Basel implementation. A possible reason for this may be the pressure from shareholders to advance fewer loans to mitigate the pressure of Basel II implementation. Among the bank-level control variables, CAR has a significant adverse relationship with credit risk, which reveals that banks with more capital have a lower credit risk. Basel II also concludes the same thing.

These findings are similar to those of (Anani & Owusu, 2023; Hakenes & Schnabel, 2011; Jiang & Yuan, 2022; Sarkar et al., 2019). Other bank-level control variables, for instance AGE and SIZE, have no significant influence on banks' credit risk in models M1–M4. On the other hand, LIQ has a significant negative effect on credit risk in all models, which suggests that banks with more liquidity prefer low credit risk. With regard to macroeconomic variables, GDP has a positive effect on banks' credit risk, while INFR has a negative significant effect, which suggests that when GDP increases, banks' lending also increases. As a result, more loans become non-performing, so credit risk increases, while increases in inflation reduce such risk. The p-value of lagged risk is significant for models M1-M4, which suggests that credit risk demonstrates persistency.

Variables	M1	M2	M3	M4	M5	M6	M7	M8
NPL(-1)	0.154***	0.143***	0.147***	0.160***	0.117***	0.126***	0.138***	0.279^{***}
	(21.43)	(23.62)	(24.25)	(20.36	(55.86)	(17.06)	(11.74)	(7.83)
BR	-0.093**			-0.077**			-0.423***	
552	(-7.13)			(-5.81)	0.4.64000		(-18.55)	
BR ²					0.164***		0.153	
CDD		0.000***	0 511***		(17.37)		(18.22)	0 5 4 5 ***
CKK		-0.929	-0.544					-0.545
CDD ²		(-10.77)	(-7.76)			Q 2Q0***		(-10.47)
CKK						0.300 (10.10)		(12 32)
SIZE	-0.003	0.001	0.014	-0.003	-0.001**	0.001**	-0.006**	(12.32) -0.001**
SILL	(-2.36)	(0.47)	(-2.25)	(-2, 12)	(-1, 1)	(-3.16)	(-2452)	(-1253)
CAR	-0.004***	-0.005^{***}	-0.004^{***}	004***	-0.005***	-0.005^{***}	-0.005^{***}	-0.005^{***}
	(-272.57)	(-155.31)	(-9.62)	(-70.50)	(-136.98)	(-57.20)	(-37.15)	(-31.41)
LIQ	-0.113***	-0.141***	-0.096***	133***	-0.148***	-0.159***	-0.151***	-0.169***
	(-13.54)	(-6.06)	(-100.38)	(-15.79)	(-17.01)	(-26.34)	(-28.65)	(-25.05)
AGE	0.002	0.002	-0.002	0.002	0.002***	0.002***	0.002^{***}	0.003***
	(10.89)	(10.8)	(-1.21)	(19.66)	(18.33)	(23.84)	(18.62)	(22.52)
BASEL			-0.016^{***}	-0.002			-0.012^{***}	-0.014^{***}
			(-7.45)	(1.93)			(-8.23)	(-7.35)
BASEL*BR							-0.003^{***}	
							(-8.55)	de de de
BASEL*CRR								-0.240^{***}
25 5							0 0 1 0 ***	(-16.14)
GDP	0.001	0.001	0.001	-0.001	0.017***	0.001	0.019	0.002
	(-0.65)	(1.16)	(2.72)	(-6.65)	(16.31)	(0.34)	(6.38)	(2.81)
INFR	-0.003***	-0.003	-0.002	-0.003***	-0.001***	-0.002	-0.001	-0.003
Constant	(-8.41)	(-1/.62)	(-10.46)	(-9.69)	(-5.32)	(-9.78)	(-7.64)	(-9.80)
Constant	(11.70)	0.258^{*}	(2,00)	(6.22)	(15, 15)	(10, 80)	0.307	(2, 27)
$\mathbf{AD}(1)$ m volue	(11.79)	(13.48)	(2.09)	(0.33)	(15.15)	(19.89)	(4023)	(3.37)
AR(1) p-value AR(2) p value	506	0.000	0.000	0.000	687	0.000	0.000 601	0.000
AR(2) p-value Sargan Test (n	0.000	0.741	.407	0.303	.087	0.302	0.000	0.751
value)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Hansen test	0.875	0 476	0.880	0 989	0 954	0.876	0.959	0 988
(p-value)	5.675	5.175	5.000	5.202	5.201	5.675	5.202	5.200
No. of Obs.	494	494	494	494	494	494	494	494
No. of Banks	33	33	33	33	33	33	33	33
NT					1. 0. 5			1.100

Table 5. Impact of MP on bank risk-taking behavior (When Risk= NPL).

Notes: The table displays the two-step difference GMM estimate results of Equation (1). NPL, CAR, and MP are endogenous variables. The NPL (-1) is lagged NPL. BR and CRR are the measures of MP, while BASEL is a dummy variable. SIZE, AGE, CAR, and LIQ are bank level control variables, whereas GDP and INFR are macro-economic control variables. In all Models, credit risk (RISK) is the dependent variable which is measured by NPL to TL. The values enclosed in parenthesis represent t-statistics. *, ** and *** indicate the significance level at the 10%, 5%, & 1% respectively.

Data Science in Finance and Economics

We further extended the base line and extended models by including the squared value of BR (M5) and CRR (M6) to observe the non-linear association between MP and risk, and by including the product of BASEL and BR (M7), and BASEL and CRR (M8) to show the joint effect of BASEL and MP, as bank risk is controlled by Basel capital requirements. From models M5 and M6, it can be concluded that there is a significant positive association between credit risks and the square of BR and CRR, which suggests that a non-linear association exists between MP and risk. It can be stated that when BR and CRR increase, credit risk decreases, but after a certain point, further increments in BR and CRR mean the credit risk also increases. From models M7 and M8, it can be said that the joint effect of BASEL and MP is significant and negative; that is, banks' risk-taking behavior is primarily influenced by both BASEL capital requirements and MP in Bangladesh. For models M5-M8, risk is persistent, which means banks with high credit risk in the previous period also face more risk in the current period. The table also reveals that large banks face low credit risk. This is due to their ability for diversification in investment because of their high level of assets. Capital and liquidity have a negative and significant effect on bank risk in all models (M1–M8). Regarding bank age, old banks face more credit risk. The effects of GDP and INFR are similar to the base line and extended models. The p-value of AR (2) indicates that there is no second order serial correlation, while the p-value of the Sargan test suggests that the over-identification restriction is valid.

5.4.2. The influence of MP and Basel II on risk-taking behavior when default risk (stability) is considered

Table 6 shows the results of the two-step difference GMM applied to equations (1) and (2). In models M9, M10, M11, and M12, the default risk measured by LNZ, that is, the logarithm of the Z-score, is employed as the dependent variable as a proxy for bank risk. In models M9 and M12, BR is the measure of MP, while in models M10 and M11 CRR is used. The effect of Basel II is shown in models M11 and M12. Table 6 reveals a positive and statistically significant relationship between LNZ and LNZ (-1), indicating that there was persistence in the stability of banks during the study period, and implying that if a bank is stable in the earlier period, it will also be stable in the present year. According to both MP measures, the table reveals a negative and statistically significant relationship between LNZ and MP, showing that when BR and CRR increase, LNZ decreases, which means that the default risk also increases.

Basel II has no significant impact on bank stability, which suggests that the stability of the economic sector prior to and following the implementation of Basel II is indifferent. Among the bank-level control variables, CAR has a significant positive impact on stability, suggesting that banks with a high level of CAR are more stable than those with a low level. Table 6 also shows that the effects of AGE, LIQ and SIZE on bank stability are not statistically significant, except in three cases. The effect of liquidity on stability is positive and significant when BR is used as a proxy for MP, while the effect is not statistically significant when CRR is used as a proxy. This suggests that banks with more liquidity are more stable than those with low liquidity. With regard to macro-economic variables, GDP and IFR have a negative and significant impact on LNZ, which indicates that as GDP and INFR increase, LNZ decreases, which means that default risk increases.

In accordance with the literature, the models were extended further to observe the non-linear association between MP and bank risk (M13 and M14) and to include the joint effect of Basel II and MP (M15 and M16). as it was introduced to mitigate risk in the banking sector. The results demonstrate a nonlinear connection between MP and the default risk of banks. The relationship

between the squared value of BR and CRR and banks' default risk is positive and significant, which suggests that when BR and CRR increase LNZ first decreases, but after further increments it will in fact increase. This suggests that, due to MP, when BR and CRR increase, banks' default risk first increases, but later decreases [see Models M15 and M16].

Variables M9 M10 M11 M12 M13 M14 M15	M16
LNZ(-1) 0.523*** 0.546*** 0.547*** 0.543*** 0.531*** 0.548*** 0.532***	0.545***
(24.93) (34.56) (26.25) (24.69) (26.01) (34.29) (18.07) (0.0214333) (0.0213333) (0.021333) (0.021333) (0.021333) (0.021333) (0.021333)	(21.62)
$BR = -0.931^{+++} = -0.761^{+++} = -0.316^{+++} = -0.316^{+++} = -0.316^{+++} = -0.316^{+++} = -0.316^{++++} = -0.316^{++++} = -0.316^{++++} = -0.316^{++++} = -0.316^{++++} = -0.316^{++++} = -0.316^{+++++} = -0.316^{+++++} = -0.316^{+++++} = -0.316^{+++++} = -0.316^{+++++} = -0.316^{++++++} = -0.316^{++++++} = -0.316^{+++++++} = -0.316^{++++++++++++++++++++++++++++++++++++$	
(-4.0) (-4.03) (2.00) (2.00)	
$DK^{-} 0.080^{++++} 0.080^{++++} 0.080^{++++} 0.080^{++++} 0.080^{++++} 0.080^{+++++} 0.080^{+++++} 0.080^{+++++} 0.080^{+++++} 0.080^{++++++} 0.080^{++++++} 0.080^{++++++++} 0.080^{+++++++} 0.080^{++++++++++} 0.080^{++++++++++++} 0.080^{+++++++++++++++++++++++++++++++++++$	
$(3.42) \qquad (^{-4.13})$	-15 318
(-2.5) (-5.24)	-13.318
(-5.5) (-5.54) CDD2 40.915***	(-0.42)
(-3.20)	(-0.39)
SIZE 0.031 0.064 0.005 -0.001 0.027 $0.064***$ -0.017	0.049
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(1.43)
(1.49) (4.62) (0.2) (0.03) (1.23) (4.62) (0.04)	(1.45)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(6.37)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	(0.37)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(-0.48)
(2.57) (0.70) (1.34) (1.74) (1.71) (0.70) (1.75)	(0.48)
AGE 0.003 0.002° 0.002° 0.002° 0.002° 0.002° (-2.81) (-2.20) (-2.82) (-2.01) (-2.20) (-2.01)	(-1, 1)
(-2.01) (-2.59) (-2.59) (-2.02) (-2.91) (-2.59) (-5.01)	(-1.1) 1 212**
$\begin{array}{cccc} DASEL & 0.175 & 0.090 & -0.125^{++++} \\ (4.00) & (2.16) & (.0.54) \end{array}$	-1.212^{++}
(4.99) (5.10) (-0.34) (0.020**	(-1.04)
BASEL*BK 0.029***	
(0.83)	26 674**
BASEL*UKK	20.0/4**
	(1.83)
$GDP = -0.001^{*} -0.01/^{***} -0.014^{*} -0.011 = 0.063^{***} -0.01/^{***} = 0.050^{**}$	-0.022
(08) (-2.7) (-1.79) (-1.15) (2.73) (-2.7) (1.97)	(-1./6)
INFR 0.028^{***} 0.029^{***} 0.029^{***} 0.031^{***} -0.020^{***} 0.029^{***} -0.036	-0.018**
(3.91) (-5.17) (-4.73) (-5.08) (-3.45) (-5.17) (-1.34)	(-1.94)
Constant $2.147 \times 0.543 \times 1.339 \times 2.250 \times 2.290 \times 0.429 \times 0.749$	
(4.22) (2.46) (3.69) (5.21) (4.57) (2.46) (1.26)	1.503***
	(1.73)
AR(1) p- 0.000 0.000 0.000 0.000 0.000 0.000 0.000	0.000
value	
AR(2) p- 0.382 0.587 0.784 0.686 0.583 0.683 0.585	0.665
value	
Sargan Test 0.000 0.000 0.000 0.000 0.000 0.000 0.000	0.000
(p-value)	
Hansen test 0.782 0.786 0.711 0.710 0.741 0.786 0.669	0.689
(p-value)	
No. of Obs. 494 494 494 494 494 494 494	494
No. of Banks 33 33 33 33 33 33 33	33
Notes: The table displays the two-step difference GMM estimation results of Equation (1) and (2). The en	ndogenous
variables are MP, CAR, and LNZ. BR and CRR are the proxies for MP. Bank level control factors include SI	ZE, CAR,

Table 6. Impact of MP and BASEL II on bank risk-taking behavior (When Risk= LNZ).

Notes: The table displays the two-step difference GMM estimation results of Equation (1) and (2). The endogenous variables are MP, CAR, and LNZ. BR and CRR are the proxies for MP. Bank level control factors include SIZE, CAR, LIQ, and AGE. BASEL is a dummy proxy for BASEL II. Macro-economic control variables include GDP, and IFR. The values enclosed in parenthesis represent t-statistics. *,**, and *** indicate the significance level at 10%, 5%, & 1% respectively.

The table also shows that when the non-linear model and the joint effect of BASEL and MP are considered, the effect of SIZE is significant and negative, which implies that large banks hold lower

LNZ, suggesting that they have more default risk. For models M9-M16, CAR and LIQ are not significant factors in banks' LNZ (default risk). In the extended models (M13–M16), bank AGE has a significant positive impact on LNZ, suggesting that older banks tend to have more LNZ and are consequently more stable. The table also reveals that BASEL and MP have a significant and positive joint impact on bank stability, suggesting that BASEL is not alone in influencing bank stability. Therefore, control mechanisms such as MP are required; through MP, banks hold more LNZ than in the period before Basel II. The effect of macroeconomic variables in the extended models shows consistent results, in line with those in the base line and base line extended models. The p-value of AR (2) shows that there is no serial correlation of the second order, while the p-value of the Sargan test suggests that the over-identification restriction is valid.

In summary, it can be concluded that MP has a nonlinear relationship with bank risk in Bangladesh. This suggests that when BR and CRR increase through MP, bank credit risk and default risk first decrease, but later increase. Lagged risk indicates that banks' risks were persistent during the study period. Among the bank level control variables, the results also reveal that banks with greater capital face low credit risk and that large banks face a lower level of credit and default risk. Apart from Model 1, liquidity is positively important in all the other models. This suggests that banks with more liquidity face a lower level of risk, whereas liquidity is not an important factor in bank stability. Likewise, the outcomes of the macroeconomic variables are striking. In accordance with the results of Changjun Zheng et al. (2017) and Zhou and Tewari (2019), but contrary to those of Brana et al. (2019) and Zins and Weill (2017), higher GGDP raises bank risk because banks spend more during economic expansion, resulting in greater risk. Similarly, this would have negative effects on the Z-score. Due to the availability of funds in debtors' hands, the INFR is substantially and negatively correlated with bank credit risk across all credit risk models. Therefore, inflation reduces the risk of private commercial banks and increases their solvency. In our primary models, we excluded the effect of Basel implementation and the findings are unremarkable. This is because of the absence of the Basel effect, as Basel is a capital-based regulation aimed at mitigating bank risk. But when Basel was included in the extended model, the results are surprising, and it is also evident from the findings that Basel II and MP jointly influence banks' risk-taking behavior in Bangladesh.

5.4.3. The nonlinear relationship between MP and risk

Some studies demonstrate a nonlinear and quadratic relationship between MP and risk. Incorporating square and quadratic terms of MP into the regressions reveals this fact. We included a squared term of BR and CRR in the extended models, with the results presented in Tables 5 and 6, to consider the nonlinear relationship between MP and risk (stability). After allowing for nonlinearity and quadratic terms, all the models demonstrate a statistically significant relationship between MP and risk when risk is the dependent variable. The tables show that the squared term of both MP measures exhibits a positive association with credit risk and stability (see Tables 5 and 6). This suggests that when BR and CRR increase, banks first face a lower credit risk, but later this increases. The association between the squared value of BR and CRR and banks' default risk is positive and significant, which suggests that there is a U-shaped relationship between the square of MP measures and bank risk, and that when BR and CRR increase, LNZ first decreases, but after further increments it increases, but later decreases [see Model M15 and M16]. This conclusion supports

the U-shaped association proposed by (Haris et al., 2020). Interestingly, the squared value of BR and CRR was included in the extended models (M7, M8, M15 and M16), and both credit risk and stability exhibited persistency. All the bank-level control variables and macroeconomic variables have a significant influence on bank risk-taking behavior in Bangladesh.

5.5. Robustness and analysis

Variables	M17	M18	M19	M20
LLP (-1)	0.912***(293.44)	0.919***(395)	0.918***(377.3)	0.908***(240.65)
Spread	-0.279***(-16.86)	-0.416***(-33.15)	$1.480^{**}(6.67)$	-2.699***(-6.46)
Spread ²			-22.502***(-8.49)	12.930***(3.01)
SIZE2	-0.008***(-15.91)	-0.005****(-13.47)	-0.005***(-11.31)	$-0.008^{***}(-17.19)$
EQTA	-0.008***(-33.49)	-0.008***(-33.09)	$-0.008^{**}(-25.98)$	-0.007***(-20.6)
LIQ	-0.032***(-8.61)	-0.028***(-12.14)	-0.029***(-10.06)	-0.033***(-9.99)
AGE	0.001***(20.49)	0.001****(13.78)	0.001***(12.89)	0.001***(24.71)
BASEL	0.007***(12.27)			-0.070 * * * (-10.14)
Spread*BASEL				1.494***(11.36)
GDP	0.001***(3.95)	0.001***(11.34)	0.001***(6.2)	0.001***(4.54)
INFR	0.001***(8.91)	0.001***(7.39)	0.001***(9.49)	-0.001 * * * (-2.79)
Constant	0.060***(12.32)	0.051***(14.7)	0.014***(2.63)	0.160***(12.78)
AR(1) p-value	0.005	0.006	0.007	0.005
AR(2) p-value	0.952	0.934	0.945	0.966
Sargan Test (p-value)	0.000	0.000	0.000	0.000
Hansen test	0.507	0.592	0.622	0.434
(p-value)				
No. of Obs.	527	527	527	527
No. of Banks	33	33	33	33

Notes: The table displays the two-step difference GMM estimation results of Equation (1) and (2). The endogenous variables are MP, EQTA, and LLP. Spread is the proxy for MP. Bank level control variables include SIZE, CAR, LIQ, and AGE. BASEL is a dummy proxy for BASEL II. Macro-economic control variables include GDP, and INFR. The values enclosed in parenthesis represent t-statistics. *,**, and *** indicate the level of significance at 10%, 5%, and 1% respectively.

In order to assess the robustness of the regression results, the baseline and extended models were modified. We first substituted credit risk with loan loss provision instead of the non-performing loan ratio (LLP) as an alternative risk measure. Second, we used spread as a measure of MP instead of CRR and BR, and third, CAR was replaced with the ratio of shareholders' equity to total assets (EQTTA), as an alternative measure of capitalization. Finally, SIZE2 was used instead of SIZE, measured by the number of bank branches. The regression results are shown in Table 7. All the estimates support our previous findings. We observe that risk is persistent, and that the interaction between MP and LLP is significant and negative, suggesting that an increase in spread reduces bank risk. The squared value of spread exhibits a U-shaped relationship between MP and risk, which supports the previous findings. Therefore, our key findings are similar to previous findings, with a few exceptions; for instance, banks tended to have a greater liquidity risk during the pre-2010 period before Basel II than during the post-2010 period, whereas there was no such correlation for credit risk. The necessity to maximize investor capital in order to produce more revenue than its costs during the Basel II period before 2010, which was supported by depositors, is a valid reason for this. Similarly, the application of EQTA reveals the possible outcome for all the models. Similar to Table 6, the coefficients of the MP variables are virtually identical here. Although the primary regression findings are different from the coefficient of EQTA for this risk model, the difference is not statistically significant. This evidence indicates that one macroeconomic variable alone does not have a significant impact on bank risk. Almost all the main variables, and all the control variables, and macroeconomic variables exhibit similar signs of associations, which suggests that our model is robust when alternative measures of the different variables were used.

6. Concluding remarks and policy implications

This study has investigated the association between bank risk and MP in developing a country, with consideration of the role of Basel II. The most up-to-date comprehensive data from Bangladesh covering the period 2002 to 2021 were used. To account for endogeneity between MP, capital, risk, heteroskedasticity, serial correlation, and cross section dependency, difference GMM was employed in the analysis. The appropriate tests suggest that difference GMM was suitable in our case. The main results show that there was a significant impact of MP on banks' risk-taking, and that the relationship between MP and bank risk is U-shaped, suggesting that banks face low risk when BR and CRR increase, but after a certain point and further increments in BR and CRR, the risk increases. The inclusion of Basel II in the model makes it fit, indicating that MP and Basel II capital regulation jointly effect bank risk. Regarding the effect of Basel II, the results suggest that banks took fewer risks than before its implementation, meaning it was successful in encouraging banks to reduce risk. The findings also reveal that MP, together with Basel II, motivates banks to reduce credit risk and to increase stability, Basel II should be followed by a control mechanism such as MP.

With regard to the bank level control variables, bank size has a negative significant influence on both credit risk and stability, suggesting that large banks take on less risk than smaller ones, and that the stability of large banks is lower than that of small banks. As expected, capitalized banks face low credit risk, which is the ultimate objective of Basel capital regulations. Surprisingly, it is suggested that the level of risk taken by a bank does not have a notable impact on its overall stability. In addition, more liquid banks face a low level of credit risk, but liquidity is not an influential factor in bank stability. The positive and significant association between risk and bank age indicates that older banks face more credit risk than younger ones, and that the effect of liquidity on bank stability is not statistically significant. Finally, regarding the macroeconomic variables, the results reveal that economic growth has a positive and significant impact on banks' credit risk, but that inflation has negative significant impact. The effect of GDP on bank stability is negative and significant, while the impact of INFR is positive and significant.

The findings point to extremely beneficial implications for stakeholders such as policymakers, regulators, stockholders, financial analysts, and researchers. We have demonstrated that MP and capital-based Basel regulations are crucial in controlling credit risk. With effective MP and capital-based regulations, banks can be incentivized to take fewer risks, thus maintaining greater capital stability. Second, because Basel II has the potential to affect both banks' internal management policies and market factors, banks should evaluate its multiple implications. Third, Basel II requirements should be enforced and priority given to large banks with inadequate capitalization. Finally, regulators are shown to have a lack of supervisory control over the risk-taking behavior of banks, as the results indicate a clear relationship between market capitalization and risk. Therefore, the study indicates that there is evidence to support the fact that expansionary money supply

stimulates bank lending, while a contractionary monetary policy has the opposite effect. This suggests that changes in money supply can influence bank lending and real sector economic activity in developing countries. Additionally, the findings reveal that high monetary policy rates discourage bank lending in such countries. To overcome these limitations, policymakers need to develop the financial system by removing financial inflexibilities and establishing quality institutional settings. This study also highlights the importance of meeting the minimum capital adequacy ratio, as this is a crucial determinant of bank lending in developing countries such as Bangladesh. Finally, this study shows that a high inflation rate discourages lending in such countries due to the uncertainty and asymmetric lending behavior created by high and volatile inflation.

Future research could be conducted which considers the impact of corporate governance mechanisms on the relationship between MP and banks' risk taking. Research could also include, for example, the influences of ownership structure, ownership concentration, institutions' shareholding, and sponsors' shareholding on the association between bank risk and MP. Further research could also be conducted by considering cross-country data and the effects of Basel III, which is now in force in the banking sector of Bangladesh as well as in other countries.

Acknowledgment

The authors express gratitude to the editor and the two anonymous reviewers for their valuable feedback that improved the quality of this paper. All authors have read and agreed to the published version of the manuscript.

Funding

The research received no external funding.

Institutional review board statement

Ethical review and approval were waived for this study because the research does not deal with vulnerable groups or sensitive issues.

Use of AI tools declaration

The authors declare they have not used Artificial Intelligence (AI) tools in the creation of this article.

Conflicts of interest

The authors declare no conflict of interest.

References

Acharya VV, Imbierowicz B, Steffen S, et al. (2020) Does the lack of financial stability impair the transmission of monetary policy? J Financ Econ 138: 342–365. https://doi.org/10.1016/j.jfineco.2020.06.011

- Agur I, Demertzis M (2010) Monetary policy and excessive bank risk taking. 271. Available at SSRN http://dx.doi.org/10.2139/ssrn.1950167
- Aiyar S, Calomiris CW, Hooley J, et al. (2014) The international transmission of bank capital requirements: Evidence from the UK. J Financ Econo 113: 368–382. https://doi.org/10.1016/j.jfineco.2014.05.003
- Aiyar S, Calomiris CW, Wieladek T (2016) How does credit supply respond to monetary policy and bank minimum capital requirements? *Eur Econ Rev* 82: 142–165. https://doi.org/10.1016/j.jfineco.2014.05.003
- Akhter N (2023) Determinants of commercial bank's non-performing loans in Bangladesh: An
empirical evidence. Cogent Econ Financ 11: 2194128.
https://doi.org/10.1080/23322039.2023.2194128
- Al-Tamimi HAH (2008) Implementing Basel II: an investigation of the UAE banks' Basel II preparations. *J Financ Regul Compl* 16: 173–187. https://doi.org/10.1108/13581980810869814
- Altavilla C, Canova F, Ciccarelli M (2020) Mending the broken link: Heterogeneous bank lending rates and monetary policy pass-through. *J Monetary Econ* 110: 81–98. https://doi.org/10.1016/j.jmoneco.2019.01.001
- Altunbas Y, Gambacorta L, Marques-Ibanez D (2012) Do bank characteristics influence the effect of monetary policy on bank risk? *Econ Lett* 117: 220–222. https://doi.org/10.1016/j.econlet.2012.04.106
- Anani M, Owusu F (2023) Regulatory capital and bank risk-resilience amid the Covid-19 pandemic: How are the Basel reforms faring? *Financ Res Lett* 52: 103591. https://doi.org/10.1016/j.frl.2022.103591
- Anwar CJ, Suhendra I, Purwanda E, et al. (2023) Investigating the relationship between monetary policy, macro-prudential policy and credit risk in Indonesia banking industry. *Heliyon* 9. https://doi.org/10.1016/j.heliyon.2023.e18229
- Arellano M, Bond S (1991) Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *Rev Econ Stud* 58: 277–297. https://doi.org/10.2307/2297968
- Arellano M, Bover O (1995) Another look at the instrumental variable estimation of errorcomponents models. *J Econometrics* 68: 29–51. https://doi.org/10.1016/0304-4076(94)01642-d
- Avery RB, Berger AN (1991) Risk-based capital and deposit insurance reform. *J Bank Financ* 15: 847–874. https://doi.org/10.1016/0378-4266(91)90103-s
- Bashir U, Yugang Y, Hussain M (2020) Role of bank heterogeneity and market structure in transmitting monetary policy via bank lending channel: empirical evidence from Chinese banking sector. *Post-Communist Econ* 32: 1038–1061. https://doi.org/10.1080/14631377.2019.1705082
- Bekaert G, Hoerova M, Duca ML (2013) Risk, uncertainty and monetary policy. *J Monetary Econ* 60: 771–788. https://doi.org/10.1016/j.jmoneco.2013.06.003
- Berger AN, Bouwman CH (2017) Bank liquidity creation, monetary policy, and financial crises. J *Financ Stabil* 30: 139–155. https://doi.org/10.1016/j.jfs.2017.05.001
- Bernanke BS (2000) Japanese monetary policy: a case of self-induced paralysis? *Japan's financial crisis and its parallels to US experience*, 149–166. Available from: https://api.semanticscholar.org/CorpusID:18313611.

- Blundell R, Bond S, Windmeijer F (2001) *Estimation in dynamic panel data models: improving on the performance of the standard GMM estimator*, 15. Emerald Group Publishing Limited. https://doi.org/10.1920/wp.ifs.2000.0012
- Blunden T (2005) Global consumer initiatives: will Basel II work? Consum Policy Rev 15: 229.
- Bond SR, Hoeffler A, Temple JR (2001) GMM estimation of empirical growth models. *Available at SSRN 290522*. https://doi.org/10.2139/ssrn.253866
- Borio C, Zhu H (2012) Capital regulation, risk-taking and monetary policy: a missing link in the transmission mechanism? *J Financ Stabil* 8: 236–251. https://doi.org/10.1016/j.jfs.2011.12.003
- Brana S, Campmas A, Lapteacru I (2019) (Un) Conventional monetary policy and bank risk-taking: A nonlinear relationship. *Econ Modell* 81: 576–593. https://doi.org/10.1016/j.econmod.2018.07.005
- Bui DT, Nguyen CP, Su TD (2021) Asymmetric impacts of monetary policy and business cycles on bank risk-taking: evidence from Emerging Asian markets. J Econ Asymmetries 24: e00221. https://doi.org/10.1016/j.jeca.2021.e00221
- Chami R, Cosimano TF (2010) Monetary policy with a touch of Basel. *J Econ Bus* 62: 161–175. https://doi.org/10.1016/j.jeconbus.2009.12.001
- Chen M, Wu J, Jeon BN, et al. (2017) Monetary policy and bank risk-taking: Evidence from emerging economies. *Emerg Mark Rev* 31: 116–140. https://doi.org/10.1016/j.ememar.2017.04.001
- Chen TY, Kao LJ (2014) Dividend changes and information about future profitability: an application of difference GMM. *Applied Econ Lett* 21: 505–508. https://doi.org/10.1080/13504851.2013.870648
- Čihák M, Schaeck K (2007) *How well do aggregate bank ratios identify banking problems?* International Monetary Fund Washington, DC. https://doi.org/10.5089/9781451868388.001
- Das Gupta A, Sarker N, Rifat Rahman M (2021) Relationship among cost of financial intermediation, risk, and efficiency: Empirical evidence from Bangladeshi commercial banks. *Cogent Econ Financ* 9: 1967575. https://doi.org/10.1080/23322039.2021.1967575
- de Moraes CO, de Mendonça HF (2019) Bank's risk measures and monetary policy: Evidence from a large emerging economy. *N Am J Econ Financ* 49: 121–132. https://doi.org/10.1016/j.najef.2019.04.002
- de Moraes CO, Montes GC, Antunes JAP (2016) How does capital regulation react to monetary policy? New evidence on the risk-taking channel. *Econ Model* 56: 177–186. https://doi.org/10.1016/j.econmod.2016.03.025
- Delis MD, Hasan I, Mylonidis N (2017) The risk-taking channel of monetary policy in the US: Evidence from corporate loan data. *J Money Credit Bank* 49: 187–213. https://doi.org/10.1111/jmcb.12372
- Dell'Ariccia G, Laeven L, Suarez GA (2017) Bank leverage and monetary policy's risk-taking channel: evidence from the United States. *J Financ* 72: 613–654. https://doi.org/10.1111/jofi.12467
- Demsetz RS, Strahan PE (1997) Diversification, size, and risk at bank holding companies. *J Money Credit Bank* 300–313. https://doi.org/10.2307/2953695
- Djatche MJN (2019) Re-exploring the nexus between monetary policy and banks' risk-taking. *Econ Model* 82: 294–307. https://doi.org/10.1016/j.econmod.2019.01.016

- ElBannan MA (2015) Do consolidation and foreign ownership affect bank risk taking in an emerging economy? An empirical investigation. *Manag Financ* 41: 874–907. https://doi.org/10.1108/mf-12-2013-0342
- Fu B, Luo D (2021) Monetary policy uncertainty and bank leverage: Evidence from China. *Econ Lett* 203: 109866. https://doi.org/10.1016/j.econlet.2021.109866
- Furlong FT, Keeley MC (1989) Capital regulation and bank risk-taking: A note. *J Bank Financ* 13: 883–891. https://doi.org/10.1016/0378-4266(89)90008-3
- Ge X, Liu Y, Zhuang J (2023) Monetary policy uncertainty, market structure and bank risk-taking: Evidence from China. *Financ Res Lett* 52: 103599. https://doi.org/10.1016/j.frl.2022.103599
- Geng Z, Zhai X (2013) Monetary policy instruments and bank risks in China. Int J Asian Bus Inf Manag (IJABIM) 4: 57–71. https://doi.org/10.4018/jabim.2013040105
- Gondwe S, Gwatidzo T, Mahonye N (2022) Bank regulation and risk-taking in sub-Sahara Africa. J Financ Regul Compl (ahead-of-print). https://doi.org/10.1108/jfrc-12-2021-0104
- Gopalakrishnan B, Jacob J, Mohapatra S (2021) Risk-sensitive Basel regulations and firms' access to credit: Direct and indirect effects. J Bank Financ 126: 106101. https://doi.org/10.1016/j.jbankfin.2021.106101
- Gropp R, Heider F (2007) What can corporate finance say about banks' capital structures. *Eur Central Bank Work Pap.* Available from: https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp1096.pdf.
- Hakenes H, Schnabel I (2011) Bank size and risk-taking under Basel II. *J Bank Financ* 35: 1436–1449. https://doi.org/10.1016/j.jbankfin.2010.10.031
- Haque F (2019) Ownership, regulation and bank risk-taking: evidence from the Middle East and North Africa (MENA) region. *Corp Gov-Int J Bus Soc* 19: 23–43. https://doi.org/10.1108/cg-07-2017-0135
- Haris M, Tan Y, Malik A, et al. (2020) A study on the impact of capitalization on the profitability of banks in emerging markets: A case of Pakistan. J Risk Financ Manag 13: 217. https://doi.org/10.3390/jrfm13090217
- Hussain M, Bashir U, Bilal AR (2021) Effect of monetary policy on bank risk: does market structure matter? *Int J Emerg Mark* 16: 696–725. https://doi.org/10.1108/ijoem-09-2019-0674
- Jiang H, Yuan C (2022) Monetary policy, capital regulation and bank risk-taking: Evidence from China. *J Asian Econ* 82: 101512. https://doi.org/10.1016/j.asieco.2022.101512
- Jung A (2023) Are monetary policy shocks causal to bank health? Evidence from the euro area. J Macroecon 75: 103494. https://doi.org/10.1016/j.jmacro.2022.103494
- Kabundi A, De Simone FN (2020) Monetary policy and systemic risk-taking in the euro area banking sector. *Econ Model* 91: 736–758. https://doi.org/10.1016/j.econmod.2019.10.020
- Kennedy P (2008) A guide to econometrics. John Wiley & Sons. Available from: https://masonlec.org/site/rte_uploads/files/Econometrics%20Book%20-%20Intro,%20Ch%201%20and%202.pdf
- Khan HH, Ahmad RB, Gee CS (2016) Bank competition and monetary policy transmission through the bank lending channel: Evidence from ASEAN. *Int Rev Econ Financ* 44: 19–39. https://doi.org/10.1016/j.iref.2016.03.003
- Khan S (2022) The impact of capital structure on bank performance in emerging markets: Empirical evidence from GCC countries. *Financ Int Q* 18: 56–65. https://doi.org/10.2478/fiqf-2022-0005

- Mahrous SN, Samak N, Abdelsalam MAM (2020) The effect of monetary policy on credit risk: evidence from the MENA region countries. *Rev Econ Polit Sci* 5: 289–304. https://doi.org/10.1108/reps-07-2019-0099
- Malovaná S, Kolcunová D, Brož V (2019) Does monetary policy influence banks' risk weights under the internal ratings-based approach? *Econ Syst* 43: 100689. https://doi.org/10.1016/j.ecosys.2018.10.003
- Mateev M, Moudud-Ul-Huq S, Nasr T (2021) Capital regulation and market competition in the MENA region: policy implications for banking sector stability during COVID-19 Pandemic. *Glob Bus Rev* 09721509211064442. https://doi.org/10.1177/09721509211064442
- Mateev M, Sahyouni A, Tariq MU (2023) Bank regulation, ownership and risk taking behavior in the MENA region: policy implications for banks in emerging economies. *Rev Manag Sci* 17: 287– 338. https://doi.org/10.1007/s11846-022-00529-5
- Matthys T, Meuleman E, Vander Vennet R (2020) Unconventional monetary policy and bank risk taking. *J Int Money Financ* 109: 102233. https://doi.org/10.1016/j.jimonfin.2020.102233
- Mohammed N, Ismail AG, Muhammad J (2015) Evidence on market concentration in Malaysian dual banking system. *Procedia-Soc Behav Sci* 172: 169–176. https://doi.org/10.1016/j.sbspro.2015.01.351
- Moudud-Ul-Huq S (2021) Does bank competition matter for performance and risk-taking? Empirical evidence from BRICS countries. *Int J Emerg Mark* 16: 409–447. https://doi.org/10.1108/ijoem-03-2019-0197
- Naceur SB, Marton K, Roulet C (2018) Basel III and bank-lending: Evidence from the United States and Europe. *J Financ Stabil* 39: 1–27. https://doi.org/10.1016/j.jfs.2018.08.002
- Naqvi H, Pungaliya R (2023) Bank size and the transmission of monetary policy: Revisiting the lending channel. *J Bank Financ* 146: 106688. https://doi.org/10.1016/j.jbankfin.2022.106688
- Ngambou Djatche MJ (2022) Monetary policy, prudential policy and bank's risk-taking: A literature review. *J Econ Surv* 36: 1559–1590. https://doi.org/10.1111/joes.12485
- Nguyen VHT, Boateng A (2015) An analysis of involuntary excess reserves, monetary policy and risk-taking behaviour of Chinese banks. *Int Rev Financ Anal* 37: 63–72. https://doi.org/10.1016/j.irfa.2014.11.013
- Ogunmola GA, Chien F, Chau KY, et al. (2022) The Influence of Capital Requirement of Basel III Adoption on Banks' Operating Efficiency: Evidence from US Banks. *J Cent Bank Theor Pr* 11: 5–26. https://doi.org/10.2478/jcbtp-2022-0011
- Olivero MP, Li Y, Jeon BN (2011) Consolidation in banking and the lending channel of monetary transmission: Evidence from Asia and Latin America. *J Int Money Financ* 30: 1034–1054. https://doi.org/10.1016/j.jimonfin.2011.06.006
- Özşuca EA, Akbostancı E (2016) An empirical analysis of the risk-taking channel of monetary policy in Turkey. *Emerg Mark Financ Trade* 52: 589–609. https://doi.org/10.1080/1540496x.2015.1047300
- Rahman MM, Hamid MK, Khan MAM (2015) Determinants of bank profitability: Empirical evidence from Bangladesh. *Int J Bus Manag* 10: 135. https://doi.org/10.5539/ijbm.v10n8p135
- Ridwan R, Mayapada AG (2022) Does sharia governance influence corporate social responsibility disclosure in Indonesia Islamic banks? In (12: 299–318): Taylor & Francis. https://doi.org/10.1080/20430795.2020.1749819

- Roy R, Bashar OH, Bhattacharya PS (2023) The cross-industry effects of monetary policy: New
evidence from Bangladesh. Econ Model 106479.
https://doi.org/10.1016/j.econmod.2023.106479
- Rubio M, Carrasco-Gallego JA (2016) The new financial regulation in Basel III and monetary policy: A macroprudential approach. *J Financ Stabil* 26: 294–305. https://doi.org/10.1016/j.jfs.2016.07.012
- Saif-Alyousfi AY, Saha A (2021) Determinants of banks' risk-taking behavior, stability and profitability: Evidence from GCC countries. *Int J Islamic Middle* 14: 874–907. https://doi.org/10.1108/imefm-03-2019-0129
- Sarkar S, Sensarma R (2019) Risk-taking channel of monetary policy: Evidence from Indian banking. *Margin: J Appl Econ Res* 13: 1–20. https://doi.org/10.1177/0973801018800088
- Sarkar S, Sensarma R, Sharma D (2019) The relationship between risk, capital and efficiency in Indian banking: Does ownership matter? *J Financ Econ Policy* 11: 218–231. https://doi.org/10.1108/jfep-05-2018-0074
- Siddique A, Khan MA, Khan Z (2021) The effect of credit risk management and bank-specific factors on the financial performance of the South Asian commercial banks. *Asian J Account Res* 7: 182–194. https://doi.org/10.1108/ajar-08-2020-0071
- VanHoose DD (2008) Bank capital regulation, economic stability, and monetary policy: what does the academic literature tell us? *Atlantic Econ J* 36: 1–14. https://doi.org/10.1007/s11293-007-9100-z
- Veeramoothoo S, Hammoudeh S (2022) Impact of Basel III liquidity regulations on US Bank performance in different conditional profitability spectrums. *N Am J Econ Financ* 63: 101826. https://doi.org/10.1016/j.najef.2022.101826
- Wu J, Yan Y, Chen M, et al. (2022) Monetary policy, economic uncertainty and bank risk: Crosscountry evidence. J Int Money Financ 122: 102580. https://doi.org/10.1016/j.jimonfin.2021.102580
- Zheng C, Moudud-Ul-Huq S, Rahman M, et al. (2017) The effects of ownership structure on banks' capital and risk-taking behavior: empirical evidence from developing country. *Res Int Bus Financ* 42: 404–421. https://doi.org/10.1016/j.ribaf.2017.07.035
- Zheng C, Moudud-Ul-Huq S, Rahman MM, et al. (2017) Does the ownership structure matter for banks' capital regulation and risk-taking behavior? Empirical evidence from a developing country. *Res Int Bus Financ* 42: 404–421. https://doi.org/10.1016/j.ribaf.2017.07.035
- Zhou S, Tewari DD (2019) Shadow banking, risk-taking and monetary policy in emerging economies: A panel cointegration approach. *Cogent Econ Financ* 7: 1636508. https://doi.org/10.1080/23322039.2019.1636508
- Zins A, Weill L (2017) Islamic banking and risk: The impact of Basel II. *Econ Model* 64: 626–637. https://doi.org/10.1016/j.econmod.2017.05.001



© 2023 the Author(s), licensee AIMS Press. This is an open access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0).